Author’s response to reviews

Title: Data for Improvement and Clinical Excellence: Protocol for an Audit with Feedback Intervention in Long Term Care

Authors:

Anne E Sales (anne.sales@ualberta.ca)
Corinne Schalm (cschalm@shepherdscare.org)

Version: 4 Date: 7 October 2010

Author’s response to reviews:

Have addressed formatting changes requested-- see below:

Formatting changes to be made:
-----------------------------

Major revisions (we require the author to make these changes)

Authors’ contributions: Please indicate that all authors read and approved the final manuscript.-- this has been added

Additional files: all additional files must be mentioned in the text in numerical order, or removed from the system. You do not mention any of the additional files in the manuscript..If you would like to include an appendix, it must be provided in the manuscript and must consist of text only. Please remember to delete the additional files section from the manuscript-- appendices A through D are mentioned in the text and added as additional files. The remaining two additional files were for review purposes and have been deleted.

Figure 1: is mentioned in the text but you do not have a figure uploaded to the submissions system. -- I have deleted the reference in the text as I don't think this figure added sufficiently to the paper to justify inclusion.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

References: please provide the full citation details for reference 26. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmccgenet/ifora/#references-- have done this.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.-- have done this and I did not find any typographic
errors.